Author's response to reviews

Title: Investigating the purpose of an online discussion group for health professionals: a case example from forensic occupational therapy.

Authors:

Crystal Dielman Grass (crystal.dieleman.grass@dal.ca)
Edward AS Duncan (edward.duncan@stir.ac.uk)

Version: 4 Date: 27 May 2013

Author's response to reviews: see over
Ms. Flora Mae Calumpita/Mr. Stephano Bonacina
BioMed Central Editorial

March 2013

MS: 1026903735804542

Ms. Calumpita,

Reviews of our paper titled ‘Investigating the utility of an online discussion group for health professionals: A case example from forensic occupational therapy’ were received by email on May 16, 2013. Revisions based on recommendations from the reviewers have been made to the manuscript (see attached) and are summarized below.

Reviewer report: Carolyn Murray

Minor Revisions

1. We have removed all use of the first person in the limitations and recommendations section.

Respectfully,

Crystal Dieleman  Edward Duncan
crystal.dieleman@dal.ca  edward.duncan@stir.ac.uk